Author's response to reviews

Title: Anaphylactic reaction associated with Ranitidine in a patient with acute pancreatitis: a Case report.

Authors:

Ulfin Rethnam (ulfinr@yahoo.com)
Rajam S Yesupalan (ajeesh2000@yahoo.co.uk)

Version: 5 Date: 28 August 2007

Author's response to reviews:

To
The Editor,
Journal of Medical Case Reports

Sir,

Thank you for accepting our case report for publication. We would like to submit the revised case report titled “Anaphylactic reaction associated with Ranitidine in a patient with acute pancreatitis: a case report” for publication in your esteemed journal. The manuscript has been formatted as per the suggestions.

1. The title has been re-formatted and changed into sentence case.
2. The title has been aligned to the left.
3. The authors' qualifications have been removed.
4. The corresponding author has been denoted as * in the title and the telephone number has been removed.
5. Keywords have been removed from the manuscript.
6. The Competing interest declaration has been changed as advised.
7. The author declaration has been modified.

Please do let us know regarding any further modifications to the manuscript.

Thanking you,

Yours truly,

Mr Ulfin Rethnam